Author's response to reviews

Title: Access to Health care and Employment Status of People with disabilities in South India, the SIDE Study (South India Disability Evidence) Study: A Case Control Study

Authors:

Murthy Venkata S Gudlavalleti (murthy.gvs@iiphh.org)
Neena John (neena.sushil@iiphh.org)
Komal Allagh (komal.allagh@iiphh.org)
Jayanthi Sagar (jayanthi.sagar@iiphh.org)
Sureshkumar Kamalakannan (suresh.kumar@iiphh.org)
Srikrishna S Ramachandra (srikrishna.sr@iiphh.org)

Version: 2
Date: 26 May 2014

Author's response to reviews:

1. Line numbers have now been incorporated and all page breaks in the word document have been removed.
2. In the abstract, ‘findings’ has been changed to ‘results’.
3. Author’s contributions have been formatted as per the suggested guidelines and all authors are referred to by their initials in the revised document.
4. Funding details have been incorporated into the acknowledgements section.